Author's response to reviews

Title: Maternal blood manganese level and birth weight: a MOCEH birth cohort study

Authors:

Jin-Hee Eum (hhsmvp@skku.edu)
Hae-Kwan Cheong (hkcheong@skku.edu)
Eun-Hee Ha (eunheeha@ewha.ac.kr)
Mina Ha (minaha@dku.edu)
Yangho Kim (yanghokm@nuri.net)
Yun-Chul Hong (ychong1@snu.ac.kr)
Hyesook Park (hpark@ewha.ac.kr)
Nam-Soo Chang (nschang@ewha.ac.kr)

Version: 2
Date: 1 February 2014

Author's response to reviews:

Dear reviewers,

I am not sure whether the revised manuscript is a "clean" version or traceable version. Please beware that the lines and pages on the response paper is based on the "traceable" version. If clean version is provided by the editors, indices may not indicate a right position. I apologize for the confusion, if any.